Author's response to reviews

Title: Pharmacotherapy of elderly patients in everyday anthroposophic medical practice: a prospective, multicenter observational study

Authors:

Elke Jeschke (ejeschke@havelhoehe.de)
Thomas Ostermann (thomaso@uni-wh.de)
Horst C Vollmar (horst.vollmar@dzne.de)
Matthias Kröz (mkroez@havelhoehe.de)
Angelina Bockelbrink (angelina.bockelbrink@charite.de)
Claudia M Witt (claudia.witt@charite.de)
Stefan N Willich (stefan.willich@charite.de)
Harald Matthes (hmatthes@havelhoehe.de)

Version: 3 Date: 16 June 2010

Author's response to reviews: see over
Dear Editor,

Please find enclosed our revised manuscript “Pharmacotherapy of elderly patients in everyday anthroposophic medical practice - a prospective, multicenter observational study”.

We are grateful for the principle acceptation of our manuscript. Below you will find a point-by-point list of replies to the reviewers’ concerns.

Thank you again for your time and consideration!

Yours sincerely,
Elke Jescke

Response to the comments

Referee 1
Minor Essential Revisions

Referee 1: The authors should provide the statistics (X2, df, and p-value) for Hosmer-Lemeshow goodness-of-fit.

Author: Done as requested (Page 9, para 2).

Referee 2
Referee 2: The authors have done a good job responding to my comments and suggestions, particularly with respect to better defining how drugs were categorized for the purposes of their research. Other than some minor style and grammatical issues, it is my belief that the manuscript is ready for publication.

Author: The revised version of the manuscript has now been copy-edited and proofread by a native speaker.